%% Math 131, Fall, 2003
\documentclass[12pt]{article}
\usepackage{latexsym,amsmath,epic,eepic}

\pagestyle{empty}

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{0in}
%% \setlength{\topmargin}{-.5in}
\setlength{\textheight}{8.25in}

\def\ds{\displaystyle}
\def\vp{$\vphantom{\big|}$}

\def\R{\textbf{R}}
\def\Item{\item{$\Box$}\quad}

\def\be{\begin{enumerate}}
\def\ee{\end{enumerate}}

\def\bd{\begin{description}}
\def\ed{\end{description}}

\def\Answer#1{$\phantom{.}$\hfill
        #1\quad\fbox{$\phantom{\Big|Answer}$}} 

\newcommand{\MyAnswer}[2]{$\phantom{.}$\hfill
        #1\quad\fbox{$\vphantom{\Big|}#2$}}

\newcommand{\soln}{\paragraph{Solution}}
\newcommand{\sep}{{\medskip\hrule\bigskip}}
\newcommand{\blank}{\underline{$\phantom{A blank}$}}

\begin{document}

\noindent\textbf{Name}:\ \vbox{\hrule width 3in}


\begin{center}
\textbf{Holy Cross College, Fall Semester, 2003} \\
\textbf{Math 131, Quiz \#4, Friday, October 3} \\
Professor Hwang \\
\end{center}

\be
\item Each of the questions below concerns the function~$f$
  \textbf{whose derivative is shown}. Circle the correct answer(s) as
  appropriate. \\ 
  (Two points for each correct answer, $-2$~for each incorrect answer.)

\begin{center}
\input{quiz04.eepic}
\end{center}

  \bd
    \item{(a)} At which point(s) does~$f$ have horizontal tangent? \qquad
      $B$\quad $C$\quad $D$\quad $E$\quad $F$

      \vskip 0pt plus 0.2fil
  
    \item{(b)} On which interval(s) is~$f$ increasing? \qquad
      $[A,C]$\qquad $[B,D]$ \qquad $[E,G]$

      \vskip 0pt plus 0.2fil
  
    \item{(c)} Is~$f$ \textbf{increasing} or \textbf{decreasing}
      at~$A$? 

      \vskip 0pt plus 0.2fil
  
    \item{(d)} Does the graph of~$f$ look like \quad$\smile$\quad or
      \quad$\frown$\quad at~$D$? 

      \vskip 0pt plus 0.2fil
  
  \ed

\ee
\eject

\end{document}
